

Author's response to reviews

Title: Risk factors for recurrent injurious falls that require hospitalization for older adults with dementia: a population based study

Authors:

Lynn Meuleners (l.meuleners@curtin.edu.au)

Michelle Fraser (m.fraser@curtin.edu.au)

Max Bulsara (Max.Bulsara@nd.edu.au)

Kyle Chow (C.Chow@curtin.edu.au)

Jonathon Ng (jonathon.ng@uwa.edu.au)

Version: 2 Date: 21 May 2016

Author's response to reviews:

1. Please submit the final copy with no track changes visible.

Final copy has been attached

2. Please add a 'Declarations' heading to your manuscript after the 'conclusions' section. This declarations section should contain all of the sub-sections according to the submission guidelines: <http://bmcneurol.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article>

Declarations section has been added.